REVIEW HISTORY

Manuscript Title: A Comprehensive review of interpersonal pragmatics: theoretical developments and applications in the Chinese context

Submitted: 28 May, 2025 Accepted: 17 July, 2025 Published: 23 July, 2025

Corresponding Author: Hongqi Ji Email: jihongqi1105@163.com

Doi: http://dx.doi.org/10.19044/llc.v11no1a6

PEER REVIEW

Reviewer 1: Fabio Ciambella Reviewer 2: Barbara Cappuzzo